Author's response to reviews

Title: Retino-protective effect of Bucida buceras against oxidative stress induced by H2O2 in human retinal pigment epithelial cells line.

Authors:

Simon B Iloki Assanga (ilokiasb@yahoo.com)
Lidianys M Lewis Luján (lidianys1@yahoo.es)
Daniela Fernandez Ángulo (dfernandez@rubiopharma.com)
Armida A Gil Salido (andreagil@libaf.com)
Claudia L Lara Espinoza (claudialara@libaf.com)
Jose L Rubio Pino (medicinabiologica@rubiopharma.com)

Version: 3
Date: 22 December 2014

Author's response to reviews: see over
1. Competing interests: Please include a 'Competing interests' section between the Conclusions and Authors' contributions. If there are none to declare, please write 'The authors declare that they have no competing interests'. Please consider the following questions and include a declaration of competing interests in your manuscript:

   In the manuscript was declared no conflict of interest by the authors.

2. Authors' contributions: Please include an Authors' contributions section before the Acknowledgements and Reference list.

   In the manuscript was incorporated the section authors contributions

3. Please ensure that your Abstract is formatted into sub-sections, as per our formatting guidelines.

   The abstract was formatted according formatting guidelines

4. Thank you for including an Acknowledgements section in your manuscript. However, we would ask you to move this section to its correct location at the end of your manuscript.

   The Acknowledgements section was located at the end of the manuscript.

5. Thank you for uploading your figures into our submission system. However, we would ask you to please remove the legends from these figures. Figure legends should only be included as a list in your manuscript document.

   The legends from figures were removed of the manuscript document.

Thank you for receives our article. We make these changes at the moment and we hope for our suggestion in the next contact.

Best regards

Dr. Iloki Assanga